Author's response to reviews

Title: Studying the consumption and health outcomes of fiscal interventions (taxes and subsidies) on food and beverages in countries of different income classifications; a systematic review

Authors:

AMAAP Alagiyawanna (ajithalagi@yahoo.com)
Nick Townsend (nicholas.townsend@dph.ox.ac.uk)
Oli Mytton (otm21@medschl.cam.ac.uk)
Pete Scarborough (peter.scarborough@dph.ox.ac.uk)
Nia Roberts (nia.roberts@bodleian.ox.ac.uk)
Mike Rayner (mike.rayner@dph.ox.ac.uk)

Version: 3
Date: 23 August 2015

Author's response to reviews: see over
Author’s response to review

Studying the consumption and health outcomes of fiscal interventions (taxes and subsidies) on food and beverages in countries of different income classifications; a systematic review

AMAAP Alagiyawanna¹, Nick Townsend², Oli Mytton³, Pete Scarborough², Nia Roberts⁴, Mike Rayner²

Corresponding author: 
AMAAP Alagiyawanna, Ministry of Health, Health Education Bureau, No 2, Kynsey Road, Colombo, Sri Lanka

ajithalagi@yahoo.com

Date: 22nd August 2015

Author’s response to reviews: see over
To the editors,

Thank you very much for the opportunity to revise and resubmit our manuscript and thank you to the reviewers and editors for their comments. We have acted on all comments and we believe that we have substantially improved the paper. Small changes of the title was done to better reflect the remit of the review. Table 1 and 2 were reformulated according to BioMed Central guidelines.

Our point-by-point responses are provided below.

Many thanks

**AMAAP Alagiyawanna**

Replies to editors comments

<table>
<thead>
<tr>
<th>Comment</th>
<th>Authors? Contributions need to be included</th>
</tr>
</thead>
<tbody>
<tr>
<td>Response</td>
<td>Details has been added on this page 16, lines 418-430:</td>
</tr>
</tbody>
</table>

**Author Contributions**

All authors conceived the review. Wrote the first draft of the manuscript: AA. Development of search strategy: AA and NR. Contributed to the writing of the manuscript: AA, NT, PS OM and MR. selected the studies: AA, NT and OM. Extracted data: AA, NT.Contributed substantially to the methods for addressing quality of studies in the review: AA and NT. All authors read and approved the final manuscript.

**Funding**

No funding was obtained specifically for this review. AA’s position was funded by the Government of Sri Lanka. NT (006/P&C/CORE/2013/OXFSTATS), MR, and PS’s (021/P&C/Core/2010/HPRG) positions at Oxford University are funded by the BHF. OM’s position was an NIHR Academic Clinical Fellowship, his current PhD position is funded by the Welcome Trust. Nia Roberts’s is employed by the Bodleian Health Care Libraries. No
funders had a role within the review.